

September 22, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: ESPS Manuscript 5424-Review.doc).

Title: Efficacy of mosapride plus proton pump inhibitors for treatment of gastroesophageal reflux disease: a systematic review

Author: Qing Liu, Chen-Chen Feng, Er-Man Feng, Xiu-Juan Yan, Sheng-Liang Chen

Name of Journal: World Journal of Gastroenterology

ESPS Manuscript NO: 5424

The manuscript has been improved according to the suggestions of reviewers.

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer.

(1) According to the advice of the reviewer, we mainly focused on the paralleled comparison group.

(2) It's an important suggestion that although the studies that assessed the additional efficacy of mosapride in PPI resistant patients showed positive results, the poor quality of these open trials should be taken into consideration. Therefore, we cannot simply draw the conclusion that mosapride plus PPIs is of benefit to PPI-resistant GERD patients. We have discussed this part more critically, and cut short the content concerning the efficacy of mosapride in PPI-resistant patients in our revised manuscript.

(3) With regard to the title, we used the full name of proton pump inhibitors and gastroesophageal reflux disease instead of acronym to express the extent more accurately. Therefore, the word count of the title is more than 12 words.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in *the World Journal of Gastroenterology*.

Sincerely yours,

*Shengliang Chen*

Sheng-Liang Chen, MD, Professor

Department of Gastroenterology

Ren Ji Hospital, School of Medicine

Shanghai Jiao Tong University

145 Middle Shandong Road

Shanghai 200001, China

Telephone: +86-21-63200784; Fax: +86-21-63266027

E-mail: slchenmd@hotmail.com